Author's response to reviews

Title: Infant feeding practice and associated factors of HIV positive mothers attending prevention of mother to child transmission and antiretroviral therapy clinics in Gondar Town health institutions, Northwest Ethiopia

Authors:

Dagnachew F Muluye (fetenemulu24@gmail.com)
Desalegn K Woldeyohannes (desalegn01@gmail.com)
Mucheye B Gizachew (muchegiza@gmail.com)
Moges W Tiruneh (mogest4@gmail.com)

Version: 5 Date: 15 February 2012

Author's response to reviews:

We have tried to address the issues raised by the reviewer but the aim of the first issue is not clear for us “Child age must also be part of the age-relevant tables and figures”. We include those subjects with less than 2 years old child to increase the sample size though it has recall bias which is indicated in the limitation part. We just assessed the feeding patterns of infants while under six years old. Probably this issue is out of our objective. If it is understood in this way we need clarification otherwise we consider it out of the objective.

The other issues are addressed accordingly.